Author’s response to reviews

Title: Giant hydronephrosis mimicking progressive malignancy

Authors:
- Dr. Andres J. Schrader (ajschrader@gmx.de)
- Georgia Anderer (georgia.anderer@rocketmail.com)
- Rolf von Knobloch (Rolf.von-Knobloch@med.uni-marburg.de)
- Dr. Axel Heidenreich (heidenre@post.med.uni-marburg.de)
- Rainer Hofmann (R.Hofmann@mail.uni-marburg.de)

Version: 3  Date: 14 Oct 2003

To:
Enitan Sawyerr
Editorial Production Assistant
BMC Journals

Resubmission of 1365 6809 6818 0308
Marburg, October 14, 2003

Dear Enitan Sawyerr,

Thank you for your email concerning our manuscript entitled "Giant hydronephrosis mimicking progressive malignancy" (ID: 1365 6809 6818 0308).

We have now made the required formatting changes, i.e. changed the numbers of the figures and the referencing of the figures in the text.

Please do not hesitate to contact me if you have any questions regarding the manuscript.

With best wishes,

Andres Jan Schrader, M.D.
Philipps-University of Medicine
Dept. of Urology
Baldingerstrasse
D-35043 Marburg
Tel.: +49/6421-286-2514
Fax: +49/6421-286-5590
Email: ajschrader@gmx.de